Author's response to reviews

Title: It's all about relationships: A qualitative study of health researchers' perspectives of conducting interdisciplinary health research

Authors:

Kalpana M Nair (nairk@mcmaster.ca)
Lisa Dolovich (ldolovic@mcmaster.ca)
Kevin Brazil (brazilk@mcmaster.ca)
Parminder Raina (praina@mcmaster.ca)

Version: 4 Date: 8 May 2008

Author's response to reviews: see over
May 8, 2008

Dear Dr. Norton:

We are excited about having our manuscript, *It’s all about relationships: A qualitative study of health researchers’ perspectives of conducting interdisciplinary health research*, accepted for publication. As requested, the following 3 changes have been made:

1. Quotation marks have been removed from the title;
2. The author list in the manuscript is exactly as the authors are listed in the submission system; and
3. Sentence case has been used for all headings.

If you have any questions about this submission, please do not hesitate to contact me at 905-522-1155 ext. 34919 or at nairk@mcmaster.ca.

Sincerely,

Kalpana Nair, MEd MSc on behalf of

Lisa Dolovich, BScPhm PharmD MSc
Kevin Brazil, PhD
Parminder Raina, PhD

*Father Sean O’Sullivan Research Centre, St. Joseph’s Healthcare, Hamilton • Faculty of Health Sciences, McMaster University*